Author’s response to reviews

Title: Clinical and genetic risk factors for new-onset diabetes mellitus after transplantation (NODAT) in major transplant centres in Malaysia

Authors:

Rhanye Mac Guad (rhaneye@ums.edu.my)
Andrew W. Taylor-Robinson (a.taylor-robinson@cqu.edu.au)
Wu Yuan Seng (sywu@mahsa.edu.my)
Gan Siew Hua (Gan.SiewHua@monash.edu)
Nur Lisa Zaharan (nurlisazaharan@um.edu.my)
Roma Choudhury Basu (romacb@um.edu.my)
Constance Liew Sat Lin (constance.liew@ums.edu.my)
Wan Ahmad Hafiz Wan Md Adnan (wahafiz@um.edu.my)

Version: 2 Date: 28 Aug 2020

Author’s response to reviews:

Dear esteemed Editor and reviewers,

We thank you for your time and effort in reviewing our manuscript and for providing constructive feedback to improve our manuscript. Please find below our point-to-point responses to the editors/reviewer’s comments:

Editor Comment 1: Please move your statement in 'Consent for publication' to appear in the 'Ethics Approval and Consent to Participate' section.

Author response: We have moved the statement in 'Consent for publication' to the 'Ethics Approval and Consent to Participate' section. (Please refer to Ethics approval and consent to participate section, line 309-312, page 13).

Editor Comment 2: The 'Consent for publication' section covers manuscripts that contain any individual person’s data in any form (including individual details, images or videos), for which consent to publish must be obtained from that person, or in the case of children, their parent or legal guardian. If your manuscript does not contain any individual person’s data, please state “Not applicable” in this section.
Author response: As the manuscript does not contain any individual person’s data in any form (including individual details, images or videos), we have written “Not applicable” in 'Consent for publication' section. (Please refer 'Consent for publication' section, line 313-314, page 13).

Editor Comment 3: At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Author response: We have uploaded a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough, or text in different colours. All relevant tables/figures/additional files were also a clean version. Figures (and additional files) remain uploaded as separate files. All figures, tables and additional/supplementary files are cited within the text.